Author's response to reviews

Title: A presenting with Abnormal Fingernail Beds Following Carbon Monoxide Poisoning A Case Report and Review of the Literature

Authors:

Mehrangiz Hatami (mehrangizhatami@yahoo.com)
Frederick Naftolin (Frederick.Naftolin@nyumc.org)
Massoud A Khatamee (frfbaby@msn.com)

Version: 2 Date: 4 February 2013

Author's response to reviews:

Dear Dr. Michael Kidd,

This is revised manuscript entitled “A presenting with Abnormal Fingernail Beds Following Carbon Monoxide Poisoning A Case Report and Review of the Literature” for publication in “The Journal of Medical Case Reports”. This manuscript is has not been published previously, and is not under consideration for publication elsewhere.